**Author’s response to reviews**

**Title:** Cardiovascular risk factors and memory decline in middle-aged and older adults: The English Longitudinal Study of Ageing

**Authors:**

Beatriz Olaya (beatriz.olaya@pssjd.org)

Maria Victoria Moneta (mvictoria.moneta@pssjd.org)

Martin Bobak (m.bobak@ucl.ac.uk)

Josep Maria Haro (jmharo@pssjd.org)

Panayotes Demakakos (p.demakakos@ucl.ac.uk)

**Version:** 1 **Date:** 10 Dec 2018

**Author’s response to reviews:**

**Please see letter to reviewers submitted as a supplementary material, since it includes tables and figures.**